
Mauna Loa Solar Observatory Observer's Log

Fri Jun 27 16:55:24 GMT 1997

Year: 97 Doy: 178

Observer: koon

Fri Jun 27 16:56:45 GMT 1997 CHIP Startup--Initializing new tape

Fri Jun 27 16:59:36 GMT 1997 dPMon Start Patrol

Fri Jun 27 16:59:39 GMT 1997 CHIP CHIP Start Patrol

Fri Jun 27 17:00:55 GMT 1997 dPMon Flat Fri Jun 27 17:01:52 GMT 1997 dPMon End Flat

WEATHER COMMENT: Fri Jun 27 16:55:30 GMT 1997 Scattered cirrus clouds, thick lower clouds to the East, wind=10 mph from

the SSE, temp=50 F.

Fri Jun 27 17:03:34 GMT 1997 CHIP Bias Fri Jun 27 17:04:18 GMT 1997 CHIP End Bias Fri Jun 27 17:04:28 GMT 1997 CHIP Water

Fri Jun 27 17:04:59 GMT 1997 CHIP End Water

MKIII COMMENT: Fri Jun 27 17:31:01 GMT 1997

I can't get the DEC computer to boot up. It starts up like normal up to the point where I toggle the Boot switch, usually the prompt "QUESTION?" appears on the printer terminal and the monitor screen asks for the

password, but I'm not getting either of those prompts.

Fri Jun 27 18:02:53 GMT 1997 dPMon Flat

Fri Jun 27 18:03:01 GMT 1997 CHIP Bias

Fri Jun 27 18:03:51 GMT 1997 CHIP End Bias

Fri Jun 27 18:04:00 GMT 1997 CHIP Water

Fri Jun 27 18:04:32 GMT 1997 CHIP End Water

WEATHER COMMENT: Fri Jun 27 18:17:26 GMT 1997

Those thick clouds from the East have moved over us.

Fri Jun 27 18:17:55 GMT 1997 dPMon End Patrol

CHIP COMMENT: Fri Jun 27 18:17:48 GMT 1997

Pausing due to thick clouds.

DPMON COMMENT: Fri Jun 27 18:18:01 GMT 1997

Pausing due to thick clouds.

Fri Jun 27 18:20:04 GMT 1997 CHIP CHIP End Patrol

MKIII COMMENT: Fri Jun 27 18:25:26 GMT 1997

I talked to Al Kai the DEC guy, he doesn't understand what is wrong, he'll be back on Monday to try again. The DEC computer is still down and we

still can't get any MKIII data until it is fixed completely.

MKIII COMMENT: Fri Jun 27 20:09:23 GMT 1997

Tried using backup disk #0 to boot the computer, it was write protected.

Got same results as with regular disk #0, no bootup. Took out backup

disk and reinstalled regular disk.

COMMENT: Fri Jun 27 20:20:12 GMT 1997

Since it is still cloudy and I have to finish preparing the trailer to be removed today I am shutting everything down. The tow truck will be here

any time now.

Fri Jun 27 20:21:45 GMT 1997 CHIP ending tape

COMMENT: Fri Jun 27 20:22:15 GMT 1997

Activity report:

QP: 47; 69; 83; 130; 232; 268; 280.

No coronal activity.

TAPES:

MKIII: Still have a computer problem that prevents taking data.

DPMON: P00921 CHIP: C00342

LOWL: L00461 in drive #0